Author's response to reviews

Title: A Thyroid Tumor extending to the Parapharyngeal Space

Authors:

Fikret Cetik (fikretcetik@yahoo.com.tr)
Demet Yazici (demetyazici@yahoo.com)
Aysun Uguz (aysunuz@yahoo.com)

Version: 2 Date: 18 November 2005

Author's response to reviews:

We have manipulated the case report and the discussion part of this article by opening up the surgical process of the patient and by giving additional update information about the parapharyngeal masses. We have seen at the operation that this tumor was originating from the right lobe of the thyroid gland via a narrow stalk and total thyroidectomy was performed at the end of the operation.